Author's response to reviews

Title: Prevention of morbidity in sickle cell disease - qualitative outcomes, pain and quality of life in a randomized crossover pilot trial of overnight supplementary oxygen and auto-adjusting continuous positive airways pressure (POMS 2a): study protocol for a randomized controlled trial

Authors:

Jo Howard (Jo.Howard@gstt.nhs.uk)
Baba Inusa (Baba.Inusa@gstt.nhs.uk)
Christina Liossi (C.Liossi@soton.ac.uk)
Eufemia Jacob (eufemia@sonnet.ucla.edu)
Patrick B Murphy (Patrick.b.murphy@kcl.ac.uk)
Nicholas Hart (Nicholas.Hart@gstt.nhs.uk)
Johanna C Gavlak (Johanna.Gavlak@uhs.nhs.uk)
Maria Chorozoglou (M.Chorozoglou@soton.ac.uk)
Sati Sahota (s.sahota@ucl.ac.uk)
Carol Nwosu (Carol.Nwosu@scyss.org)
Maureen Gwam (maureengwam@hotmail.co.uk)
Atul Gupta (atulgupta@nhs.net)
David C Rees (David.Rees@kcl.ac.uk)
Swee Lay Thein (sl.thein@kcl.ac.uk)
Isabel C Reading (I.C.Reading@soton.ac.uk)
Fenella J Kirkham (Fenella.Kirkham@ucl.ac.uk)
Man Ying E Cheng (M.Y.Cheng@soton.ac.uk)

Version: 2
Date: 6 February 2015

Author's response to reviews: see over
We apologise for the formatting errors which we have corrected:

1. Please ensure the title conforms to journal style for study protocol articles. The title should follow the format "__________: study protocol for a randomized controlled trial." We have modified the title: please see p.1

2. Please include the date of registration with the trial registration number at the end of the Abstract. We have added the date: please see p.2

3. Please move your funding information to an Acknowledgements section at the end of the manuscript, before the reference list. We have moved this: please see p.11

4. Please move the trial information table from the top of your manuscript to the area below the reference list. We have moved this: please see p.18

5. As your research involves humans please include a statement of ethical approval in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate. Any experimental research on humans must be in compliance with the Helsinki Declaration. If your study is taking place in multiple sites please ensure you mention all of the ethical bodies that have approved the study at the different sites. We apologise for this oversight and have added this information: please see p.5

6. Please include a list of abbreviations used and their meanings, after the Trial Status. We have added this: please see p.11

7. Please include a Competing Interests section after the list of abbreviations. If the authors have no competing interests, please state: "The authors declare that they have no competing interests." We have added this: please see p.11

8. Please include an Authors' Contributions section at the end of the manuscript, after the Competing Interests section. Each author needs to be listed individually. We suggest the following kind of format
(please use initials to refer to each author's contribution): "AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination and helped to draft the manuscript. All authors read and approved the final manuscript." We have added this please see p.11-12.

9. Please include a figure title and legend section after the reference list. We have added this-please see p.17